Author’s response to reviews

Title: Simplified scoring of the Actionable 8-item screening questionnaire for neurogenic bladder overactivity in multiple sclerosis: a comparative analysis of test performance at different cut-off points

Authors:

Peter J Jongen (ms4ri@kpnmail.nl)
Bertil F Blok (b.blok@erasmusmc.nl)
John P Heesakkers (John.Heesakkers@radboudumc.nl)
Marco Heerings (m.a.p.heerings@gmail.com)
Wim A Lemmens (Wim.Lemmens@radboudumc.nl)
Rogier Donders (Rogier.Donders@radboudumc.nl)

Version: 2 Date: 11 June 2015

Author’s response to reviews:

1. Line numbers have been added.

2. A list of abbreviations has been added:

ABSST Actionable Bladder Symptom Screening Tool–16 items version
FN False Negatives
FP False Positives
MS Multiple Sclerosis
NPV Negative Predictive Value
PPV Positive Predictive Value
RR Relapsing Remitting
SD Standard Deviation
TN True Negatives
TP True Positives
VPN Virtual Private Network

3. The abbreviations NPV and PVV have been added to the Abstract, and new version is included in the revised manuscript.